

# 

**Citation:** Orofino DHG, Passos SRL, de Oliveira RVC, Farias CVB, Leite MdFMP, Pone SM, et al. (2018) Cardiac findings in infants with in utero exposure to Zika virus- a cross sectional study. PLoS Negl Trop Dis 12(3): e0006362. https://doi. org/10.1371/journal.pntd.0006362

Editor: Walderez O. Dutra, Instituto de Ciências Biológicas, Universidade Federal de Minas Gerais, BRAZIL

Received: October 6, 2017

Accepted: March 4, 2018

Published: March 26, 2018

**Copyright:** © 2018 Orofino et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files

Funding: Sonia Regina Lambert Passos was supported by MCTI | Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq) Grants 440846/2016-1 and 310765/2016-1; supported by Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (CAPES)Grant 88881130793/2016-01 and by Universidade RESEARCH ARTICLE

# Cardiac findings in infants with in utero exposure to Zika virus- a cross sectional study

Dulce H. G. Orofino<sup>1,2®</sup>\*, Sonia R. L. Passos<sup>3®</sup>, Raquel V. C. de Oliveira<sup>3®</sup>, Carla Verona B. Farias<sup>1‡</sup>, Maria de Fatima M. P. Leite<sup>1‡</sup>, Sheila M. Pone<sup>4‡</sup>, Marcos V. da S. Pone<sup>4‡</sup>, Helena A. R. Teixeira Mendes<sup>2,3‡</sup>, Maria Elizabeth L. Moreira<sup>5‡</sup>, Karin Nielsen-Saines<sup>6‡</sup>

Department of Pediatrics, Fernandes Figueira Institute, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil,
Souza Marques School of Medicine, Rio de Janeiro, Brazil,
Laboratory of Clinical Epidemiology, Evandro Chagas National Institute of Infectology, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil,
Department of Infectology, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil,
Department of Clinical Research, Fernandes Figueira Institute, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil,
Department of Clinical Research, Fernandes Figueira Institute, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil,
Department of Medicine, Division of Infectious Diseases, University of California, Los Angeles, California, United States of America

• These authors contributed equally to this work.

‡ CVBdF, MdFMPL, SMP, MVdSP, HARTM, MELM, and KNS also contributed equally to this work. \* dulceorofino@yahoo.com.br

# Abstract

#### Background

Antenatal exposure to Zika virus (ZIKV) is related to severe neurological manifestations. A previous study in Brazil reported an increased incidence of non-severe congenital heart defects in infants with diagnosis of congenital Zika syndrome but without laboratory confirmation of ZIKV infection in the mother or infant. The objective of this study is to report echo-cardiographic (ECHO) findings in infants with laboratory confirmed antenatal exposure to ZIKV.

## Methodology

Cross sectional study of cardiologic assessments of infants born between November 2015 and January 2017 with confirmed vertical exposure to ZIKV in Rio de Janeiro, Brazil.

## Results

The study enrolled 120 children with a median age of 97 days (1 to 376 days). In utero exposure to ZIKV was confirmed in 97 children (80,8%) through positive maternal polymerase chain reaction (PCR) results during pregnancy or a positive PCR result at birth; 23 additional children (19.2%) had maternal positive PCR results during pregnancy and postnatally. Forty- eight infants (40%) had cardiac defects noted on ECHO. Thirteen infants (10.8%) had major cardiac defects (atrial septal defect, ventricular septal defect, patent ductus arteriosus). None of the defects were severe. The frequency of major defects was higher in infants whose mothers had a rash in the 2nd trimester of pregnancy, or who had altered Central Nervous System (CNS) imaging postnatally or were preterm. Estácio de Sá-Bolsa Produtividade. Helena Alencar Rosa Teixeira Mendes was supported by Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (CAPES) Grant 88887130789/2016-00. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Conclusions

Infants with *in utero* ZIKV exposure have a higher prevalence of major cardiac defects, however none were severe enough to require immediate intervention. For this reason, guidelines for performance of postnatal ECHO in this population should follow general newborn screening guidelines, which significantly reduces the burden of performing emergent fetal or neonatal ECHOs in a setting where resources are not available, such as most Brazilian municipalities.

#### Author summary

We performed cardiologic assessment of 120 infants born between November 2015 and January 2017 with confirmed vertical exposure to Zika virus (ZIKV). The diagnosis of ZIKV exposure was confirmed by PCR in maternal blood specimens and/or amniotic fluid during pregnancy and/or infant specimens including PCR of the cerebrospinal fluid (CSF) and/or urine following birth. Our study found a 10.8% incidence of structural heart defects in infants with a history of intrauterine exposure to ZIKV, a rate considerably higher than that observed in the general population. However, no patients were found to have severe structural heart defects. The presence of major defects was increased in infants whose mothers had a history of rash in the 2nd trimester, infants with altered CNS imaging and/or preterm infants. As severe heart defects requiring immediate intervention were not identified, guidelines for performance of postnatal ECHO in this population should follow the same criteria as the ones applicable to newborns in general, particularly since resources for performance of fetal and neonatal echocardiograms are not routinely available in most Brazilian municipalities

## Introduction

Zika virus (ZIKV) is an arbovirus of the Flavivirus genus like Dengue virus (DENV) and Yellow Fever virus (YFV) [1]. In response to the search for the causative agent of a "dengue-like" exanthematous disease in 2015, ZIKV was identified for the first time in Brazil by reverse transcriptase polymerase chain reaction (RT- PCR) in blood samples from patients in the Northeast region of the country [1]. Autochthonous transmission of ZIKV was confirmed in Brazil months later [2]. That same year, an increase in the number of infants with microcephaly born months after the outbreak was observed [3], thereby corroborating the possibility of congenital defects caused by vertical transmission of ZIKV. In 2016 the virus was identified by RT-PCR in tissue samples from two neonates with severe microcephaly [4] and by the detection of ZIKV in the amniotic fluid from two fetuses of women with history of rash during pregnancy [5,6,7]. Diagnosis of vertical exposure to ZIKV is therefore based on positive RT-PCR in mothers with suspected symptoms during pregnancy [5]. Congenital Zika syndrome, resulting from vertical transmission of the virus, includes microcephaly, specific findings from imaging of the central nervous system, visual and auditory deficits, and arthrogryposis [8,9,10].

Just as there are no reports, prior to 2015, of severe neurological manifestations related to vertical transmission of any Flavivirus, there are no reports of congenital heart defects in patients with confirmed vertical exposure to these viruses. A study in 2015 in Pernambuco State, Brazil, with 103 patients with diagnosis of congenital Zika syndrome but without

laboratory confirmation of ZIKV infection in mothers or infants showed an increased incidence of non-severe congenital heart defects [11].

With the increase in the number of studies ordered in mother-infant pairs with suspected or confirmed ZIKV infection in pregnancy, it becomes important to verify which tests are really necessary to promote better use of health resources in Brazil and elsewhere while still enabling optimal care for these patients [12]. During the ZIKV epidemic in Rio de Janeiro, we performed 2D echocardiogram in neonates and infants with suspected *in utero* exposure to the virus. The present study describes the cardiac findings identified in this population.

#### Methods

We performed a cross-sectional study of infants born between November 2015 and January 2017 who had confirmed vertical exposure to ZIKV and were followed at our outpatient Pediatric Infectious Disease Clinic at the Fernandes Figueira Institute (IFF-FIOCRUZ). The infants who received echocardiograms were born either at IFF or other institutions with positive ZIKV maternal PCR results during pregnancy or positive PCR results at birth. Our institution is a national research referral center for high risk pregnancies and high risk infants. For this reason, during the Zika epidemic we were the major referral center in Rio de Janeiro for cases of suspected maternal or infant ZIKV infection.

All infants had negative serology result for other congenital infectious including toxoplasmosis, rubella, cytomegalovirus, hepatitis B, C and HIV, in addition to no diagnosis of genetic syndromes potentially accompanied by congenital heart defects. None of the children had previous fetal ECHOs. The diagnosis of ZIKV exposure was confirmed by RT-PCR in a blood sample from the mother and/or in the amniotic fluid during pregnancy and/or in infant specimens including PCR of the cerebrospinal fluid (CSF) and/or urine following birth.

The following data were abstracted from medical records: maternal history of rash during pregnancy, gestational age at birth, diagnosis of congenital Zika syndrome, results of infants CNS imaging, and PCR results of maternal and infant specimens.

Minor ECHO defects were defined as: persistent foramen ovale (PFO), patent ductus arteriosus (PDA) in premature infants up to three months of age or in term infants up to 15 days of age, physiological tricuspid regurgitation, and physiological pulmonary branch stenosis in premature infants. All other ECHO defects were considered major structural defects. Congenital heart defects that required any therapeutic intervention in the first days of life were considered severe structural heart defects [13].

Infant cardiologic evaluation included a complete clinical examination and full 2D and Mmode transthoracic echocardiography with pulsed and continuous Doppler and color Doppler, performed by pediatric cardiologists from the Pediatric Cardiology Department of IFF. The equipment used was the Siemens Acuson X300 Echocardiography system, and the studies were performed without sedation. The cardiologic evaluation was performed between 1 and 376 days of age.

Data analysis was descriptive and used absolute and relative frequencies of qualitative variables, like gender, PCR results and major ECHOs defects. Additionally, the 95% confidence interval (95% CI) of relative frequencies was provided. Age in days was described by the median, minimum and maximum. The association between qualitative variables and major ECHO defects (presence or absence) was based on the comparison of absolute and relative frequencies, considering differences of at least 10 percentage points (pp) as indicative of a potential difference between patients with and without major ECHO alterations. In addition, individual data were provided on all patients with major ECHO defects. Due to the study's descriptive design and the small number of major ECHO defects, we avoid using statistical testing and we added Fisher exact test as only an additional information. The R software version 3.3.3 and the package epiDisplay were used to perform the data analyses.

The study was approved by the Institutional Review Board of the Brazilian National Institute of Infectious Diseases (INI-FIOCRUZ). All parents or guardians provided written informed consent. All data analyzed were anonymized.

#### Results

Among 186 children who met inclusion criteria (maternal and/ or infant PCR results confirming in utero ZIKV exposure) 120 were brought by their parents or guardians to the pediatric clinic for a cardiac consult and performance of an echocardiogram. Sixty-six children were excluded from the study because they did not come for an echo study. The median age among the 120 participants was 97 days (1 to 376 days) with equal distribution of boys and girls. Eigthy nine subjects (74.2%; 95%CI = 65.4–81.7) were born at term. Ninety-seven infants (80.8%; 95%CI = 72.6–87.4) had confirmation of *in utero* exposure to ZIKV by a positive maternal PCR during pregnancy or a positive PCR result at birth. A total of 23 additional children (19.2%; 95%CI = 12.6–27.6) had positive maternal PCR results during pregnancy and a positive PCR result at birth. A total of 84 infants in the cohort had microcephaly and 25 had the diagnosis of vertical exposure to ZIKV because microcephaly was diagnosed on the morphologic ultrasound during pregnancy or were born with congenital Zika syndrome.

Forty-eight infants (40%; 95%CI = 31.2-49.3) had cardiac defects noted on ECHO. A PFO was the most common finding responsible for 72.9% (35 patients, 95%CI = 58.2-84.7) of ECHO abnormalities. Infants with a PFO had a median age of 29 days, ranging from 2 to 372 days. PDA was found in 6 patients (12.5%, 95%CI = 4.7-25.2). The median age of infants with PDA was 98 days, ranging from 2 to 288 days. Ventricular septal defects (VSDs) were seen in 4 patients (4.2%, 95%CI = 0.5-14.3) all of them were small perimembranous VSDs. Atrial septal defects (ASDs) of the small *ostium secundum* type were present in three patients (6.3%, 95% CI = 0.1-17.2). Physiological tricuspid regurgitation was seen in three patients (6.3%, 95% CI = 0.1-17.2) and arterial pulmonary hypertension in one infant. No cases of altered ventricular function, pericardial effusion, or coronary anomalies were found. One additional patient presented with left ventricular hypertrophy without dysfunction which was unrelated to the mother's gestational diabetes. Two infants had more than one defect.

Of 48 infants with ECHO abnormalities, 13 infants had major structural defects noted, representing 10.8% of 120 infants (95% CI = 5.9% to 17.8%). None had severe structural cardiac defects. All infants in the group with major structural defects had positive PCR results in CSF and/or urine at birth as seen in Table 1. Six of 13 infants in the group with major defects had microcephaly and 5 had abnormal CNS imaging. (Table 1).

Major heart defects were 14% more frequent in preterm infants than in term babies, 10% higher in infants whose mothers had a history of rash in the 2nd trimester as compared to other time points or no rash, and 11%. higher in infants with altered CNS imaging tests. Despite this trend, no statistically significant association was identified. No difference in frequency of cardiac defects was found between infants with and without microcephaly (46.1% in the group with major cardiac defects and 42.9% in the group without cardiac defects). Distribution of findings in infants with normal and abnormal echocardiograms are outlined in Table 2.

#### Discussion

Our study found a 10.8% (IC 95% = 5.9% to 17.8%) prevalence of major structural heart defects in infants with a history of intrauterine exposure to ZIKV, a rate considerably higher

Patient	Sex	Age at ECHO (days)	PCR (m) <sup>a</sup>	PCR (n) <sup>b</sup>	Microcephaly	CNS imaging <sup>c</sup>	Gestational age (weeks)	2D-Echo	Rash-trimester
1	Girl	288	Positive	Positive	Yes	Positive	36	PDA <sup>d</sup>	2nd
2	Girl	220	Positive	Positive	No	Negative	40	PDA	2nd
3	Boy	132	Positive	Positive	No	Negative	40	VSD <sup>e</sup>	2nd
4	Boy	1	Positive	Positive	Yes	Positive	39	PAH <sup>f</sup>	1st
5	Girl	178	Positive	Positive	Yes	Positive	37	PDA	1st
6	Boy	2	Positive	Positive	Yes	Positive	40	VSD	1st
7	Girl	96	Positive	NA	Yes	Negative	38	ASD <sup>g</sup>	2nd
8	Girl	28	Negative	Positive	No	Positive	37	ASD	3rd
9	Girl	2	Positive	NA	Yes	Negative	35	LVH <sup>h</sup>	2nd
10	Boy	55	Positive	NA	No	Negative	39	ASD	2nd
11	Boy	91	Positive	Positive	No	Negative	39	VSD	2nd
12	Girl	7	Negative	Positive	No	Negative	38	POF <sup>i</sup> , VSD, RSPB <sup>j</sup>	No rash
13	Bov	NA <sup>k</sup>	Positive	NA	No	NA	37	VSD, POF	No rash

#### Table 1. Characteristics of 13 patients with abnormal ECHO findings.

a-Mother's PCR

b-Neonate's PCR

c-Central nervous system imaging

d-Patent ductus arteriosus

e-Ventricular septal defect

f-Pulmonar arterial hypertension

g-Atrial septal defect

h-Left ventricular hyperthrophy

i-Patent foramen oval

j-Relative stenosis of pulmonary branches

k-Not available

https://doi.org/10.1371/journal.pntd.0006362.t001

than that observed in the general population. However, no patients were found to have severe structural heart defects. These results are similar to those of a study from the state of Pernambuco, Brazil, which evaluated infants with congenital Zika syndrome but without laboratory

		Norma	1			Abnor	Abnormal			
		n	%	95%CI	95%CI		%	95%CI	95%CI	
Sex	Boys	53	49,5%	39,7%	59,4%	6	46,2%	19,2%	74,9%	
	Girls	54	50,5%	40,6%	60,3%	7	53,8%	25,1%	80,8%	
Gestational age <sup>a</sup>	Pre-term	26	24,3%	16,5%	33,5%	5	38,5%	13,9%	68,4%	
	Term	81	75,7%	66,5%	83,5%	8	61,5%	31,6%	86,1%	
Rash(trimester) <sup>a</sup>	No Rash	8	7,5%	3,3%	14,2%	2	15,4%	1,9%	45,5%	
	1st	33	30,8%	22,3%	40,5%	3	23,1%	5,0%	53,8%	
	2nd	48	44,9%	35,2%	54,8%	7	53,8%	25,1%	80,8%	
	3rd	18	16,8%	10,3%	25,3%	1	7,7%	0,2%	36,0%	
CNS imaging tests <sup>a b</sup>	Negative	73	69,5%	59,8%	78,1%	7	58,3%	27,7%	84,8%	
	Positive	32	30,5%	21,9%	40,2%	5	41,7%	15,2%	72,3%	

#### Table 2. Differences between patients with and without major ECHO findings.

<sup>a</sup> Variables with at least 10 percentage points at the difference between percentage of abnormal and normal ECHo groups.

<sup>b</sup> Central nervous system imaging tests

Note: Fisher exact test p-values: Sex = 1.000, Gestacional age = 0.317, Rash = 0.515, CNS Imaging test = 0.559.

https://doi.org/10.1371/journal.pntd.0006362.t002

confirmation of intrauterine exposure to ZIKV. That study demonstrated an incidence of 13.5% of major ECHO defects but also found no severe congenital heart defects [11].

The prevalence of major cardiac defects in our study was 10 times higher in ZIKV infants (13 cases in 120 infants) than that observed in the general population. The reported prevalence of congenital heart defects in live born infants in Atlanta and China respectively was 8.1 to 11.1/1,000 [13,14], with a 1.5/1000 prevalence of severe congenital heart defects reported in. In China and in the U.S. left-to-right shunts (VSD, ASD, and PDA) were the most frequent defects, and accounted for half of the congenital heart defects reported in Atlanta [13]. In the American study, muscular VSD was the most common defect while tetralogy of Fallot was the most common cyanotic heart defect (0.47/1,000). In the Chinese study, prevalence of all types of VSD was 3.7/1,000 and tetralogy of Fallot was also the most common cyanotic congenital heart defect. A study in Nigeria reported similar findings [15]. In Brazil, a recent study reported a prevalence of congenital heart disease of 0.5 per 1000 live births across the country. Nevertheless, this study strongly suggests that there is significant under reporting of cases of congenital cardiac diseases in the country [16]. Previous studies have described congenital heart disease rates in Brazil ranging from 1 per 1000 in live borns and 88 per 1000 in stillbirths [17]. Prior literature on the status of cardiac surgery for congenital heart disease in Brazil emphasizes the significant lack of resources for surgical procedures, with an overall country deficit estimated at 80% (i.e., 80% of infants in need of cardiac surgery in the first year of life did not receive them) [16], whereas mortality data following congenital heart surgery from the Heart Institute of the University of Sao Paulo was reported at 8% in 2016 [18]. This underscores the precarious condition in our country for adequate identification and management of serious congenital heart disease. From our data it appears that Zika-affected infants do not require emergent cardiac interventions.

Except for two infants with left ventricular hypertrophy and pulmonary arterial hypertension, we found left-to-right shunts in all infants with major cardiac defects, which is similar to that reported in previous epidemiologic studies of congenital heart disease. [13,14,15].

In our study the finding of PDA in three neonates who were less than 15 days of age was considered physiological, and the presence of a PFO in 29.2% of our subjects should be analyzed within the context of the infants' age (all subjects were less than 13 months of age). It is known that in postmortem analyses, the incidence of PFO is 27.3%, but taking age into account, a study showed an incidence of 34.3% in the first three decades of life and a progressive decrease in later decades [19]. However, these patients should be followed to assess whether there is any difference in the incidence of closure with age.

There was a trend in our study for major heart defects to be more frequently observed in infants with preterm birth, in those whose mothers had second trimester of pregnancy infection and in infants who had altered CNS imaging studies. Our findings however were not statistically significant, but this could be a function of sample size. We believe these factors merit further investigation in future Zika infant cohorts that have larger sample sizes.

Although our study found a higher prevalence of structural cardiac defects than anticipated, it did not demonstrate any severe structural heart defects in infants with *in utero* ZIKV-exposure. None of the infants needed cardiac procedures or surgery in the first days to months of life. We believe postnatal assessments performed routinely in maternity hospitals in Brazil which entail a newborn physical exam and a newborn oxygen saturation test prior to discharge from the hospital would be sufficient for identification of infants with major structural heart defects. Our findings support the recommendation that fetuses or infants with suspected or confirmed ZIKV exposure in utero should receive the same cardiac follow-up recommended for pregnant women and newborns in general.

Our study is the first to report echocardiogram findings in infants with laboratory confirmed antenatal exposure to ZIKV. Given the seasonality of ZIKV infection, it is difficult to predict whether it will be possible in the future to report on echocardiographic findings of a larger series of infants with laboratory confirmed diagnosis. It would be important however, to be able to replicate these findings in studies evaluating a larger number of subjects.

#### Conclusions

A higher frequency of major heart defects was noted in ZIKV exposed infants as compared to the general population. None of these defects, however were considered severe. In light of our findings, we believe recommendations for fetal echocardiograms in pregnant women with confirmed ZIKV infection and performance of postnatal ECHOs in their infants, should follow the same criteria as those established for the general population of mothers and infants. This entails a complete physical exam at birth and measure of oxygen saturation prior to discharge. Recommendations for emergent echocardiographic evaluations in the neonatal period for all Zika exposed infants do not appear to be warranted, as a higher prevalence of severe heart defects was not identified. This recommendation is very relevant to scenarios where performance of fetal and pediatric echocardiograms is not feasible, including most Brazilian municipalities which lack the ability to perform these studies. All patients in the study with positive ECHO findings are being monitored prospectively so that the spontaneous closure or possible complications are readily identified.

#### **Supporting information**

**S1 Checklist. Strobe checklist.** (DOCX)

#### **Author Contributions**

- **Conceptualization:** Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Maria Elizabeth L. Moreira.
- **Data curation:** Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Carla Verona B. Farias, Maria de Fatima M. P. Leite, Sheila M. Pone, Marcos V. da S. Pone, Helena A. R. Teixeira Mendes, Karin Nielsen-Saines.
- Formal analysis: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.

Funding acquisition: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.

- **Investigation:** Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Carla Verona B. Farias, Maria de Fatima M. P. Leite, Sheila M. Pone, Marcos V. da S. Pone, Helena A. R. Teixeira Mendes.
- Methodology: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Carla Verona B. Farias.
- Project administration: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Carla Verona B. Farias, Maria de Fatima M. P. Leite, Sheila M. Pone, Marcos V. da S. Pone.
- Resources: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.
- Software: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.
- Supervision: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Maria Elizabeth L. Moreira.

Validation: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.

Visualization: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Karin Nielsen-Saines.

Writing - original draft: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira.

Writing – review & editing: Dulce H. G. Orofino, Sonia R. L. Passos, Raquel V. C. de Oliveira, Carla Verona B. Farias, Maria de Fatima M. P. Leite, Sheila M. Pone, Marcos V. da S. Pone, Helena A. R. Teixeira Mendes, Maria Elizabeth L. Moreira, Karin Nielsen-Saines.

#### References

- 1. Campos GS, Bandeira AC, Sardi SI. Zika virus outbreak, Bahia, Brazil. *Emerg Infect Dis* 2015; 21:1885–6 https://doi.org/10.3201/eid2110.150847 PMID: 26401719
- Zanluca C, de Melo VC, Mosimann AL, Dos Santos GI, Dos Santos CN, Luz K. First report of autochthonous transmission of Zika virus in Brazil. *Mem Inst Oswaldo Cruz* 2015; 110:569–72. https://doi.org/10. 1590/0074-02760150192 PMID: 26061233
- Schuler-Faccini L, Ribeiro EM, Feitosa IM, Horovitz DD, Cavalcanti DP, Pessoa A et al. (2016). Possible association between Zika virus infection and microcephaly—Brazil, 2015. MMWR Morb. Mortal. Wkly. Rep. 65 59–62. https://doi.org/10.15585/mmwr.mm6503e2 PMID: 26820244
- Martines RB, Bhatnagar J, Keating MK, Silva-Flannery L, Muehlenbachs A, Gary J, Goldsmith C, Hale G, Ritter J et al. Notes from the Field: Evidence of Zika Virus Infection in Brain and Placental Tissues from Two Congenitally Infected Newborns and Two Fetal Losses, Brazil, 2015. *MMWR Morb Mortal Wkly* Rep2016; 65: 159–160 https://doi.org/10.15585/mmwr.mm6506e1 PMID: 26890059
- Brasil P, Pereira JP, Raja Gabaglia C, Damasceno L, Wakimoto M, Ribeiro Nogueira RM et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro -Preliminary Report. N Engl J Med 2016 Mar 4; Epub ahead of print https://doi.org/10.1056/NEJMoa1602412 PMID: 26943629
- 6. de Oliveira CS, da Costa Vasconcelos PF. Microcephaly and Zika virus. *J Pediatr* (Rio J) 2016; 92: 103–105 https://doi.org/10.1016/j.jped.2016.02.003 PMID: 27036749
- Sarno M, Sacramento GA, Khouri R, do Rosário MS, Costa F, Archanjo G et al. Zika Virus Infection and Stillbirths: A Case of Hydrops Fetalis, Hydranencephaly and Fetal Demise. *PLoS Negl Trop Dis* 2016; 10: e0004517 https://doi.org/10.1371/journal.pntd.0004517 PMID: 26914330
- Moore CA, Staples JE, Dobyns WB, Pessoa A, Ventura CV, Fonseca EB et al. Characterizing the Pattern of Anomalies in Congenital Zika Syndrome for Pediatric Clinicians. JAMA Pediatr. 2017 Mar 1; 171 (3):288–295. https://doi.org/10.1001/jamapediatrics.2016.3982 PMID: 27812690
- Ventura LO, Ventura CV, Lawrence L, van der Linden V, van der Linden A, Gois AL et al. Visual impairment in children with congenital Zika syndrome. *J AAPOS*. 2017 Apr 24. pii: S1091-8531(17) 30311-7. https://doi.org/10.1016/j.jaapos.2017.04.003 [Epub ahead of print] PMID: 28450178
- Russell K, Oliver SE, Lewis L, Barfield WD, Cragan J, Meaney-Delman D et al. Update: Interim Guidance for the Evaluation and Management of Infants with Possible Congenital Zika Virus Infection— United States, August 2016. MMWR Morb. Mortal. Wkly. Rep. 2016 Aug 26; 65(33):870–878. https:// doi.org/10.15585/mmwr.mm6533e2 PMID: 27559830
- Cavalcanti DD, Alves LV, Furtado GJ, Santos CC, Feitosa FG, Ribeiro MC, et al. Echocardiographic findings in infants with presumed congenital Zika syndrome: Retrospective case series study. *PLoS ONE* 12(4): e0175065. https://doi.org/10.1371/journal.pone.0175065 PMID: 28426680
- Brasil. Ministério da Saúde (MS). Diretrizes metodológicas: estudos de avaliação econômica de tecnologias em saúde. Brasília: MS; 2009. [accessed 2013 Oct. 20]. Available at: http://bvsms.saude.gov. br/bvs/publicaco-es/avaliacao\_economica\_tecnologias\_saude\_2009.pdf
- Reller MD, Strickland MJ, Riehle-Colarusso T, Mahle WT, Correa A. Prevalence of congenital heart defects in metropolitan Atlanta, 1998-2005. *J Pediatr.* 2008 Dec; 153(6):807–13. https://doi.org/10. 1016/j.jpeds.2008.05.059 Epub 2008 Jul 26. PMID: 18657826
- Qu Y, Liu X, Zhuang J, Chen G, Mai J, Guo X et al. Incidence of Congenital Heart Disease: The 9-Year Experience of the Guangdong Registry of Congenital Heart Disease, China. *PLoS One.* 2016 Jul 13; 11(7): e0159257. https://doi.org/10.1371/journal.pone.0159257 eCollection 2016. PMID: 27409588
- 15. Abdulkadir M, Abdulkadir Z. A systematic review of trends and patterns of congenital heart disease in children in Nigeria from 1964–2015. *Afr Health Sci.* 2016 Jun; 16(2):367–77. <u>https://doi.org/10.4314/ahs.v16i2.5 PMID: 27605952</u>
- Pinto Junior VC, Branco KMPC, Cavalcante RC, Carvalho Junior W, Lima JRC, de Freitas SM et al. Epidemiology of congenital heart disease in Brazil. *Brazilian Journal of Cardiovascular Surgery*, 30(2), 219–224. https://doi.org/10.5935/1678-9741.20150018 PMID: 26107454

- 17. Pedra CA, Haddad J, Pedra SF, Peirone A, Pilla CB, Marin-Neto JA. Paediatric and congenital heart disease in South America: an overview. *Heart*, 2009 Sep; 95(17):1385–92. <u>https://doi.org/10.1136/hrt.</u> 2008.152017 PMID: 19174420
- Caneo LF, Jatene MB, Riso AA, Tanamati C, Penha J, Moreira LF, Atik E, Trindade E, Stolf ANG. Evaluation of surgical treatment of congenital heart disease in patients aged above 16 years. *Arq. Bras. Cardiol.*, 98(5), 390–397. Epub March 29, 2012. <u>https://dx.doi.org/10.1590/S0066-782X2012005000030</u> PMID: 22460167
- Hagen PT, Scholz DG, Edwards WD. Incidence and size of patent foramen ovale during the first 10 decades of life: an autopsy study of 965 normal hearts. *Mayo Clin Proc* 1984; 59:17–20. PMID: 6694427